Author's response to reviews

Title: Trends in Liver Transplantation for Primary Biliary Cirrhosis in The Netherlands 1988-2008

Authors:

Edith M.M. Kuiper (e.m.m.kuiper@erasmusmc.nl)
Bettina E Hansen (b.hansen@erasmusmc.nl)
Herold J Metselaar (h.j.metselaar@erasmusmc.nl)
Robert A de Man (r.deman@erasmusmc.nl)
Els B Haagsma (e.b.haagsma@int.umcg.nl)
Bart van Hoek (bvhoek@lumc.nl)
Henk R van Buuren (h.vanbuuren@erasmusmc.nl)

Version: 4 Date: 26 October 2010

Author's response to reviews:

Question: Before we can process your manuscript further, we need you to make one additional change. Please can you include the name of the body or organisation which gave you permission to use the data in the manuscript?

Answer: The sentence was changed into: "NOTR and the liver transplantation centers gave permission to publish these data."